Author's response to reviews

Title: Increased Risk of Depression in Type 2 Diabetes Is Minimised by Sulfonylurea and Metformin Combination: A Population-Based Cohort Study

Authors:

Mark L. Wahlqvist (prof-mlw@umail.hinet.net)
Meei-Shyuan Lee (mmsl@ndmctsgh.edu.tw)
Shao-Yuan Chuang (chuangsy@nhri.org.tw)
Chih-Cheng Hsu (cch@nhri.org.tw)
Hsin-Ni Tsai (970721@nhri.org.tw)
Shu-Han Yu (shuhan229@gmail.com)
Hsing-Yi Chang (hsingyi@nhri.org.tw)

Version: 6 Date: 10 August 2012

Author's response to reviews: see over